Author’s response to reviews

Title: Gastric Outlet Obstruction from Metastatic Lobular Breast Carcinoma

Authors:

Alexander Kim (ahkim@geisinger.edu)
Matthew J Shellenberger (mjshellenberger@geisinger.edu)
Zong Ming Chen (zechen@geisinger.edu)

Version: 4 Date: 3 August 2015

Author’s response to reviews:

1. All the abbreviations were defined at least once during the manuscript.
2. Lobular breast adenocarcinoma was changed to carcinoma.
3. Histopathology section of the manuscript was edited by our pathologist.
4. Newer references were added and the conclusion was revised accordingly.
5. Differential diagnosis for gastric outlet obstruction was broaden including peptic ulcer disease.
6. Gastroenterology was changed to gastroenterology team per reviewer’s comment.